Author's response to reviews

Title: Inguinal lymph node metastases from a testicular seminoma: A rare occurrence - A case report and review of literature

Authors:

Mohamed Ismail Jr (ms18273@gmail.com)
Farouqz Zaman Jr (zaman17@hotmail.com)
Alistair Grey Jr (alistair.grey@homerton.nhs.uk)
Sohail Baithun Sn (s.i.baithun@qmul.ac.uk)
Vinod Nargund Sn (vinod.nargund@homerton.nhs.uk)
Junaid Masood Sn (junaid.masood@bartsandthelonon.nhs.uk)

Version: 2 Date: 7 December 2009

Author's response to reviews:

Dear Sirs

We would be grateful if you can accept our case presentation. we describe a rare presentation of testicular cancer.

looking forward hearing from you

Yours sincerely

M. Ismail